Author's response to reviews

Title: The intron 4c allele of the NOS3 gene is associated with ischemic stroke in African Americans

Authors:

Raji P Grewal (rgrewal@solarishs.org)
Ana VC Dutra (dutracacd@hotmail.com)
Yichu Liao (yichu.liao@gmail.com)
Hank Jua (hjua@c2b2.columbia.edu)
Nikolaos IH Papamitsakis (papamitsakis@yahoo.com)

Version: 2 Date: 5 July 2007

Author's response to reviews:

this has not yet been sent out for peer review